ARTICLE

Annals of Internal Medicine

Comorbidity Affects the Relationship Between Glycemic Control and Cardiovascular Outcomes in Diabetes

A Cohort Study

Sheldon Greenfield, MD; John Billimek, PhD; Fabio Pellegrini, MS; Monica Franciosi, MSc; Giorgia De Berardis, MSc; Antonio Nicolucci, MD; and Sherrie H. Kaplan, PhD, MPH

Background: Recent studies have shown mixed results regarding the effectiveness of intensive glucose-lowering therapy in reducing risk for cardiovascular events.

Objective: To determine whether attaining hemoglobin A_{1c} (Hb A_{1c}) targets of 6.5% or less or 7.0% or less for glycemic control at baseline provides differential benefits for patients with high versus low-to-moderate levels of comorbidity.

Design: 5-year longitudinal observational study of patients with type 2 diabetes. Patients were categorized into high and low-to-moderate comorbidity subgroups by using the Total Illness Burden Index (TIBI), a validated patient-reported measure of comorbidity.

Setting: 101 diabetes outpatient clinics and 103 general practitioners' clinics in Italy.

Patients: 2613 (83%) of 3074 patients with type 2 diabetes, sampled randomly from diabetes outpatient clinic rosters and recruited consecutively from general practitioners' clinics, who completed the baseline questionnaire.

Measurements: TIBI score, total mortality, and incident cardiovascular events. Hazard ratios (HRs) were adjusted for age and sex.

Results: Attaining an HbA_{1c} level of 6.5% or less at baseline was associated with lower 5-year incidence of cardiovascular events in

the low-to-moderate comorbidity subgroup (adjusted HR, 0.60 [95% CI, 0.42 to 0.85]; P = 0.005) but not in the high comorbidity subgroup (adjusted HR, 0.92 [CI, 0.68 to 1.25]; P = 0.61; P for subgroup by HbA_{1c} interaction = 0.048). Similarly, attaining a baseline HbA_{1c} level of 7.0% predicted fewer cardiovascular events in the low-to-moderate comorbidity subgroup (adjusted HR, 0.61 (CI, 0.44 to 0.83; P = 0.001) but not in the high comorbidity subgroup (adjusted HR, 0.88 [CI, 0.66 to 1.17]; P = 0.38; P for subgroup by HbA_{1c} interaction = 0.093).

Limitations: The observational nature of the study does not allow causal inference. The length of the data collection period was limited. Information on clinical management was not available.

Conclusion: Patients with the high levels of comorbidity common in type 2 diabetes may receive diminished cardiovascular benefit from intensive blood glucose control. Comorbidity should be considered when tailoring glucose-lowering therapy in patients with type 2 diabetes.

Primary Funding Source: Pfizer of Italy.

Ann Intern Med. 2009;151:854-860. www.annals.org For author affiliations, see end of text.

Agior professional organizations recommend that attaining a hemoglobin A1c (HbA_{1c}) value less than 7.0% may be less appropriate for patients with limited life expectancy, advanced complications, and extensive comorbidity (1–3). Evidence suggests that the benefit of intensive glucose-lowering therapy is not uniform across all patients with type 2 diabetes.

Three large randomized, controlled trials that used HbA_{1c} targets of 6.5% or lower (4–6) each found no association between intensive therapy and an overall reduction in risk for macrovascular complications. However,

See also:

Print
Editors' Notes
Editorial comment
Summary for PatientsI-54
Web-Only
CME quiz
Conversion of graphics into slides

when data from these and other trials were considered in 2 recent meta-analyses (7, 8), the investigators observed statistically significant relationships between tight glycemic control and reduced cardiovascular events.

Post hoc analyses of data from these clinical trials suggest that benefit from aggressive glycemic control may be confined to younger diabetic patients (4) and patients without previous heart disease (4, 5). Data from the 10-year posttrial follow-up (9) to the UKPDS (United Kingdom Prospective Diabetes Study) (10) also showed a reduction in cardiovascular events from intensive glucose-lowering therapy initiated in a young and healthy sample of patients with recently diagnosed type 2 diabetes.

Recent decision analyses based on UKPDS risk models suggest that, independent of age, high levels of comorbidity may diminish the benefits of achieving tight control (11), owing to the complex interplay of multiple conditions, their treatments, and their burden on patient resources (12). Among the comorbid conditions prevalent among diabetic patients, cardiovascular diseases are the most important contributors to mortality and subsequent cardiovascular events. However, additional conditions,

854 © 2009 American College of Physicians

such as chronic lung disease, may also bring functional impairment, treatment burden, and risk for adverse events and may diminish a patient's likelihood to benefit from tight control (13, 14).

In a 5-year observational study of a community-based sample of older patients with type 2 diabetes, we tested whether attaining glycemic control targets of HbA_{1c} levels of 6.5% or less or 7.0% or less at baseline provided differential benefits for patients with higher versus lower levels of comorbidity. We further compared the unique contribution of comorbid conditions with that of other risk factors (such as age or duration of diabetes) to the differential benefit from glycemic control on future cardiovascular events.

METHODS

Our study, described in detail elsewhere (15–17), was a 5-year longitudinal observational study (1999 to 2004) that examined the association of the quality of diabetes care with the incidence of cardiovascular events and mortality. Patients were followed for a median of 4.96 years (interquartile range, 3.35 to 5.00 years).

Patients

We identified medical practices in all regions of Italy and selected them according to their willingness to participate in the project. Participating practices included 101 of approximately 605 eligible diabetes outpatient clinics and 103 of approximately 1000 eligible community-based general practitioners enrolled in a nationwide network of practitioners interested in facilitating research.

We considered all patients with type 2 diabetes mellitus (fasting venous plasma glucose concentration \geq 7.8 mmol/L [140 mg/dL] on \geq 2 separate occasions or treated with antidiabetic drugs) to be eligible for the project, regardless of age, diabetes duration, or treatment. At the diabetes outpatient clinics, patients were randomly sampled from clinic rosters and stratified by patient age (<65 or \geq 65 years). We asked each diabetes outpatient clinic to recruit at least 30 patients. Community-based general practitioners consecutively enrolled only patients for whom they were primarily responsible for diabetes care, up to a maximum of 10 patients.

Clinical Measures

The main outcome of the study was incident cardiovascular events, defined as any of the following outcomes: angina, myocardial infarction, stroke, transient ischemic attack, coronary revascularization procedures, lower limb complications (claudication, ulcer, gangrene, amputation, or aortic–femoral revascularization procedures), or cardiovascular mortality. Participating physicians certified the occurrence of any cardiovascular event over the 5-year study period, on the basis of study-wide criteria. In addition, participating physicians reported the death of any study

Context

The effectiveness of glycemic control in reducing cardiovascular events among patients with type 2 diabetes is uncertain. Recent analyses of trial data suggest that the benefit of tight control may differ according to a person's age and comorbid conditions.

Contribution

This observational study of 2613 patients with type 2 diabetes in 205 practices in Italy found that tight control was associated with lower risk for cardiovascular events over 5 years in patients with low to moderate, but not high, comorbidity.

Caution

Observational data cannot prove a causal association between glycemic control and outcomes.

Implication

The relationship of glycemic control and cardiovascular outcomes differs by patients' level of comorbidity.

—The Editors

patient from any cause; this information was used to compute total mortality rates.

Participating physicians abstracted demographic and clinical data, including age, body mass index, duration of diabetes, HbA_{1c} level, lipid levels, and blood pressure (collected and entered into models as continuous variables), as well as sex, smoking status and the presence of diabetes complications (collected and entered into models as categorical variables) from clinical records and reported these data to the coordinating center at Mario Negri Sud. Because normal ranges for HbA1c varied in the different centers, the percentage change with respect to the upper normal value (actual value vs. upper limit of normal) was estimated and multiplied by 6.0 (16). Total cholesterol was used as a measure of lipid control because low-density lipoprotein levels were not routinely measured in many of the study patients. We used the last blood pressure value in the clinical record before the data collection point. Data were collected at baseline and at 6-month intervals for 5 years.

Measure of Comorbidity

We requested that all recruited patients complete the Total Illness Burden Index (TIBI) questionnaire (18–20). The TIBI, which was specifically developed for office practice populations, uses patient reports to assess the presence and severity of 8 dimensions of comorbid conditions, problems, and diseases (atherosclerotic heart disease, lung disease, congestive heart failure, arthritis, genitourinary disease, vision loss, gastrointestinal conditions, and foot disease) by using items similar to those in the traditional review of systems. We scored these responses to assess the

www.annals.org

Table 1. Baseline Patient Characteristics, by Comorbidity Level

Characteristic	Сото	P Value*	
	Low to Moderate (TIBI Score <12)	High (TIBI Score ≥12)	
Patients, n	1498	1115	
Mean age (SD), y	61.7 (10.5)	64.3 (9.5)	< 0.001
Men, %	58.3	50.2	< 0.001
Mean body mass index (SD), kg/m ²	27.5 (4.3)	28.4 (4.7)	< 0.001
Smoking status, %			< 0.001
Never smoked	43.0	46.9	
Current smoker	17.4	16.5	
Former smoker	34.5	35.3	
Unknown	5.1	1.3	
Mean diabetes duration (SD), y	9.7 (8.0)	11.9 (9.0)	< 0.001
HbA _{1c} level ≤7.0%, %	52.4	46.9	0.009
Mean HbA _{1c} level (SD), %	7.2 (1.7)	7.4 (1.7)	0.021
Mean total cholesterol level (SD)			0.002
mmol/L	5.5 (1.1)	5.6 (1.1)	
mg/dL	213 (41)	218 (42)	
Mean systolic blood pressure (SD), mm Hg	143.2 (17.6)	144.4 (18.4)	0.115
Mean diastolic blood pressure (SD), mm Hg	82.9 (8.5)	82.5 (8.6)	0.31

HbA_{1c} = hemoglobin A_{1c}; TIBI = Total Illness Burden Index.

* P values refer to the Pearson chi-square and Mann–Whitney U tests for categorical and continuous variables, respectively.

severity of the 8 dimensions and then aggregated the scores by using an algorithm that weighted each dimension according to its predicted effect on functional outcomes. We also performed analyses that used a version of the TIBI score that excluded previous cardiovascular events to examine the effects of the noncardiac components of the TIBI on future events. We refer to this version as the *noncardiovascular TIBI* score.

The TIBI can be completed and scored in office practices for use by physicians at the time of treatment and has been validated as a predictor of 3.5-year mortality (20) and health-related quality of life (18, 19).

Statistical Analysis

We conducted univariate analyses to describe patient characteristics and reported means and SDs for continuous variables and frequencies and percentages for categorical variables.

We calculated the probabilities of incident cardiovascular events by using the Kaplan–Meier method and carried out comparisons by using the log-rank test. We divided patients into 2 prespecified subgroups at a threshold TIBI score of 12, which has been demonstrated to discriminate between persons at greater and lesser risk for death (20). We defined patients with TIBI scores less than 12 as the low-to-moderate comorbidity subgroup and patients with scores of 12 or greater as the high comorbidity subgroup.

To account for the hierarchical nature of the data (patients clustered within center), and to control for possible confounding or clustering by center of variables, we used multivariate Cox proportional hazards regression models, stratified by center, to investigate whether a dichotomized TIBI score was an independent predictor of clinical outcomes. In all analyses, we expressed outcome risk in terms of hazard ratios (HRs) with 95% CIs in models adjusted for age (as a continuous variable) and sex (as a categorical variable).

To demonstrate that the threshold TIBI score of 12 differentiated patients' risk both for cardiovascular events and for total mortality in the current sample, we computed hazard ratios for relative risk for each outcome between participants with TIBI scores of 12 or greater and patients with scores less than 12. To demonstrate that any association observed between comorbidity and outcome risk was not an artifact of the selected TIBI score cut-point, we computed hazard ratios for both cardiovascular event risk and mortality risk by using the TIBI score as a continuous, independent variable. We replicated the analysis by using the "noncardiovascular TIBI score" to examine its association with future cardiovascular events and mortality.

To determine whether the benefit of attaining recommended targets for glycemic control was greater for patients with low-to-moderate or high levels of comorbidity, we tested whether baseline HbA1c levels of 6.5% or less or 7.0% or less were associated with lower incidence of cardiovascular events in each comorbidity subgroup. To examine whether any observed association between attaining glycemic control targets and reduced cardiovascular event risk was sensitive to the choice of TIBI cutoff thresholds, we replicated these analyses in subgroups defined by tertiles of TIBI scores for each target HbA1c value. We also tested whether we could detect differential benefits of attaining the glycemic control target in subgroups defined by high versus low noncardiovascular TIBI scores alone. Finally, to consider the contributions of variables that could be related to comorbidity, we tested separate models that included

856 15 December 2009 Annals of Internal Medicine Volume 151 • Number 12

interaction terms between HbA_{1c} level and age, duration of diabetes, sex, education, and income—each considered as a continuous variable (with the exception of sex, which is categorical). For each of these analyses, we tested the interaction of HbA_{1c} level with TIBI subgroup and with other patient characteristics (age, sex, duration of diabetes, education, and income) to assess differential levels of benefit associated with attaining tight control. We used SAS, version 9.1 (SAS Institute, Cary, North Carolina), for all analyses.

Role of the Funding Source

The study was funded by Pfizer of Italy. The funding source had no role in the design, conduct, analysis or reporting of the study or in the decision to submit the manuscript for publication.

RESULTS

Of the 3074 initially enrolled patients with type 2 diabetes, 2613 (83%) completed the baseline questionnaire and were included in our final analytic sample. Table 1 reports patient characteristics at baseline for the 2 comorbidity subgroups. Patients in the high comorbidity subgroup tended to be older than those in the low-tomoderate comorbidity subgroup (mean age, 64.3 vs. 61.7 years; P < 0.001), and fewer were male (50.2% vs. 58.3%; P < 0.001; they also were more likely to report never smoking (46.9% vs. 43.0%; P < 0.001) and had higher body mass indexes (28.4 vs. 27.5 kg/m²; P < 0.001), longer duration of diabetes (11.9 vs. 9.7 years; P < 0.001), and marginally higher levels of HbA1c (7.4% vs. 7.2%; P = 0.021) and total cholesterol (5.6 mmol/L [218 mg/ dL] vs. 5.5 mmol/L [213 mg/dL]; P = 0.002). During the 5-year follow-up period, 426 patients (16.3%) developed a cardiovascular event and 168 patients (6.5%) died.

To confirm the appropriateness of a TIBI threshold score of 12 to define subgroups, we modeled cardiovascular event risk and total mortality risk by TIBI level. Controlling for age, sex, smoking, body mass index, HbA_{1c} level, total cholesterol level, and blood pressure, patients in the

Table 2. Association Between TIBI and Risk for Cardiovascular Events or Death

high comorbidity group had a higher risk for cardiovascular events (HR, 1.52 [95% CI, 1.21 to 1.89]; P < 0.001) and death (HR, 1.39 [CI, 0.97 to 1.99]; P = 0.074) than those in the low-to-moderate comorbidity group for the 5-year observation period (Table 2).

The association between TIBI score and clinical outcomes persisted when we analyzed TIBI as a continuous variable. After adjustment, each unit change in the continuous TIBI score was associated with a 2% increase in both cardiovascular event risk (HR, 1.02 [CI, 1.01 to 1.02]; P < 0.001) and total mortality risk (HR, 1.02 [CI, 1.00 to 1.03]; P = 0.014). In addition, when we rescored the TIBI to exclude previous cardiovascular events, patients in the highest quartile of noncardiovascular TIBI scores experienced an 89% increase in risk for incident cardiovascular events compared with those in the lowest quartile, after adjustment for age and sex (HR, 1.89 [CI, 1.39 to 2.58]; P = 0.002), and showed a marginal increase in total mortality in the 5-year period after adjustment (HR, 1.52 [CI, 0.96 to 2.40]; P = 0.082).

We then tested whether attaining an HbA_{1c} target of either 6.5% or less or 7.0% or less at baseline was associated with a lower incidence of subsequent cardiovascular events in the high (TIBI score ≥ 12) versus low-tomoderate (TIBI score 12) comorbidity subgroups during follow-up (Table 3). Patients in the low-to-moderate comorbidity subgroup experienced lower rates of incident cardiovascular events if they attained the HbA_{1c} target of 6.5% or less than if they did not (2.2 events vs. 3.8 events per 100 patient-years), with an unadjusted HR of 0.58 (CI, 0.41 to 0.82) (P = 0.002) and an adjusted HR of 0.60 (CI, 0.42 to 0.85) (P = 0.005). In the high comorbidity subgroup, cardiovascular event rates did not differ between patients who attained the HbA1c target of 6.5% or less and those who did not (4.9 events vs. 5.2 events per 100 patient-years), with an unadjusted HR of 0.93 (CI, 0.68 to 1.26) (P = 0.64) and an adjusted HR of 0.92 (CI, 0.68 to 1.25) (P = 0.61). The P value for the interaction between TIBI subgroup and HbA_{1c} level was 0.036 in the

Outcome and TIBI Score		Univ	variate Analysis		Multivariate Cox H Analysis	lierarchical
	Events, n	Patients, n	Survival at 5 y (95% CI), %	Log-Rank P Value	Hazard Ratio (95% CI)*	P Value
Total cardiovascular events						
TIBI score <12	207	1498	83.7 (81.6–85.7)	< 0.001	1.00	-
TIBI score ≥12	219	1115	76.6 (73.8–79.4)		1.52 (1.21–1.89)	<0.001
Total mortality						
TIBI score <12	74	1498	94.2 (92.9–95.5)	< 0.001	1.00	-
TIBI score ≥12	94	1115	89.9 (88.0–91.9)		1.39 (0.97–1.99)	0.074

TIBI = Total Illness Burden Index.

* Adjusted for age, sex, smoking, body mass index, baseline levels of hemoglobin A1c and total cholesterol, and blood pressure.

Table 3.	Reduction in	Risk for	Cardiovascular	Events	Associated	With	HbA _{1c}	Level	6.5%	or Less	or	7.0%	or l	Less,
by TIBI	Subgroup													

Subgroup	HbA _{1c} Target ≤6.5%									
	Cardiovascular Event Rate, n per 100 patient-years			Unadjusted A	Nodel	Adjusted Model*				
	HbA _{1c} ≤6.5%	HbA _{1c} >6.5%	Change	HR (95% CI)	P Valuet	HR (95% CI)	P Value†			
TIBI score					0.036		0.048			
<12	2.2	3.8	1.6	0.58 (0.41-0.82)‡		0.60 (0.42-0.85)‡				
≥12	4.9	5.2	0.3	0.93 (0.68–1.26)		0.92 (0.68–1.25)				
TIBI score tertile					0.32		0.34			
1st	2.3	3.1	0.7	0.76 (0.49–1.20)		0.82 (0.52-1.28)				
2nd	2.9	4.9	2.0	0.60 (0.39–0.91)		0.60 (0.39–0.91)				
3rd	4.8	5.2	0.5	0.88 (0.62–1.24)		0.86 (0.61–1.23)				

 HbA_{1c} = hemoglobin A_{1c} ; HR = hazard ratio; TIBI = Total Illness Burden Index.

* Adjusted for age and sex.

+ Value for interaction between TIBI subgroup and HbA_{1c} level. + P < 0.010.

\$ P < 0.001.

|| P < 0.050.

unadjusted model and 0.048 in the adjusted model, which suggests that the cardiovascular event risk reduction associated with attaining the HbA_{1c} target of 6.5% or less in the low-to-moderate comorbidity subgroup differed from that in the high comorbidity subgroup.

We observed a similar pattern when we examined the benefit of attaining an HbA1c level of 7.0% or less. Patients in the low-to-moderate comorbidity subgroup who attained that target had a lower incidence of cardiovascular events than those who did not (2.4 events vs. 4.1 events per 100 patient-years), with an unadjusted HR of 0.59 (CI, 0.44 to 0.81) (P < 0.001) and an adjusted HR of 0.61 (CI, 0.44 to 0.83) (P = 0.001). In the high comorbidity subgroup, cardiovascular event rates did not differ between patients who attained an HbA1c level of 7.0% or less and those who did not (4.8 events vs. 5.4 events per 100 patient-years), with an unadjusted HR of 0.88 (CI, 0.66 to 1.17) (P = 0.38) and an adjusted HR of 0.86 (CI, 0.64 to 1.14) (P = 0.30). The P value for the interaction of TIBI subgroup and HbA_{1c} level was 0.061 in the unadjusted model and 0.093 in the adjusted model.

Our results from replication of these analyses in subgroups defined by TIBI score tertiles (**Table 3**) further suggest that not all patients experience equivalent benefit from attaining tight glycemic control. Patients in the highest tertile experienced similar rates of cardiovascular events whether they attained the HbA_{1c} target of 6.5% or less or did not (4.8 events vs. 5.2 events per 100 patient-years), with an unadjusted HR of 0.88 (CI, 0.62 to 1.24) (P =0.46) and an adjusted HR of 0.86 (CI, 0.61 to 1.23) (P =0.41). Patients in the second tertile who attained the HbA_{1c} target of 6.5% or less had lower rates of cardiovascular events than those who did not (2.9 events vs. 4.9 events per 100 patient-years), with an unadjusted HR of 0.60 (CI, 0.39 to 0.91) (P = 0.016) and an adjusted HR of 0.60 (CI, 0.39 to 0.91) (P = 0.017). Patients in the

858 15 December 2009 Annals of Internal Medicine Volume 151 • Number 12

fore had low cardiovascular event rates in the 5-year period, also experienced similar cardiovascular event rates whether they attained the HbA_{1c} target of 6.5% or less or did not (2.3 events vs. 3.1 events per 100 patient-years), with an unadjusted HR of 0.76 (CI, 0.49 to 1.20) (P = 0.24) and an adjusted HR of 0.82 (CI, 0.52 to 1.28) (P = 0.38). The *P* value for the interaction of TIBI subgroup and HbA_{1c} level was 0.34. Results were similar when we examined the benefit of attaining an HbA_{1c} level of 7.0% or less at baseline for each of the TIBI tertile subgroups (**Table 3**).

lowest tertile, who had little or no comorbidity and there-

We found a similar pattern of results with subgroups defined using the noncardiovascular TIBI score (data not shown), but separate tests of interactions between attaining an HbA_{1c} level of 7.0% or less and other patient characteristics (age, sex, duration of diabetes, education, and income) did not suggest differential benefit for attaining tight control across different levels of these characteristics (data not shown).

DISCUSSION

Our findings support recommendations (1-3) to focus intensive glycemic therapy on younger patients with less comorbidity and to require less stringent HbA_{1c} targets for patients with extensive complications and comorbid conditions. Among patients with low-to-moderate comorbidity, we found that baseline HbA_{1c} level was associated with reduced incidence of subsequent cardiovascular events within a 5-year period. Conversely, among patients with high levels of comorbidity, we found no association between attaining HbA_{1c} targets of 6.5% or less or 7.0% or less at baseline and experiencing a cardiovascular event during the 5-year study period.

Cardiovascular Event Rate, n per 100 patient-years			Unadjusted A	Nodel	Adjusted Model*		
HbA _{1c} ≤7.0%	HbA _{1c} >7.0%	Change	HR (95% CI)	P Value†	HR (95% CI)	P Valuet	
				0.061		0.093	
2.4	4.1	1.7	0.59 (0.44–0.81)§		0.61 (0.44–0.83)‡		
4.8	5.4	0.6	0.88 (0.66–1.17)		0.86 (0.64–1.14)		
				0.36		0.35	
2.5	3.2	0.7	0.80 (0.52-1.21)		0.84 (0.55–1.27)		
3.1	5.4	2.3	0.59 (0.41–0.84)‡		0.58 (0.40-0.84)‡		
4.6	5.5	0.9	0.81 (0.58–1.12)		0.79 (0.57–1.10)		

Our findings could shed light on the observed discrepancy between the results of the 3 recent randomized, controlled trials (4-6) that included older patients with greater comorbidity and those of the meta-analysis (7, 8) that included a broader representation of all patients, specifically younger patients with less comorbidity. If older patients with substantial comorbidity are less likely to benefit from intensive glycemic control and younger patients with less comorbidity are more likely to benefit, then the "average effect" will be influenced by the proportion of study patients that represent each group.

The hypothesis-generating post hoc analyses of recent randomized clinical trials (4, 5) illustrate the need to identify a priori subgroups to avoid "averaging" effects that could yield null results. Patients in the high comorbidity subgroup in our study had similar age and comorbidity characteristics to those in a trial that showed no benefit from tight control (4). The low-to-moderate comorbidity subgroup from our study experienced benefits that paralleled those observed in post hoc analyses among patients with "no history of macrovascular disease" and those younger than 65 years in that trial (4), and in those with "no previous cardiovascular event" in another trial (5).

Our study also suggests that even among patients with lower levels of comorbidity, the benefits of attaining tight glycemic control may not be uniform in a 5-year period. Patients with TIBI scores in the lowest tertile showed no benefit from attaining HbA_{1c} targets but may have shown benefit from tight control if they had been observed for a longer interval. The UKPDS (10), with a patient sample similar to the lowest-risk subgroup in our study, did not observe significant reductions in cardiovascular event risk until 10 years after the trial (9).

The diminished potential of tight glycemic control to reduce cardiovascular events in patients with high TIBI scores is probably due to a combination of limited life expectancy and the complexities of managing these very sick patients. The association between the TIBI score and risk for death or incident cardiovascular events persisted when we rescored the TIBI to exclude items that assessed previous cardiac disease. These findings suggest that noncardiac comorbid conditions, such as pulmonary dysfunction, gastrointestinal disease, and arthritis, may independently diminish a patient's potential to benefit from intensive glycemic control.

Our study has limitations. First, this is an observational study. We cannot establish causal links between high levels of comorbidity and diminished benefit from tight glycemic control. We also did not have information on clinical management during the 5-year observational period. However, these results lend support to the importance of comorbidity in identifying which patients may benefit from attaining intensive glycemic control. Evidence for this conclusion would be strengthened by a prospective, randomized, controlled trial designed to study the benefits of intensive glycemic control with a priori specified subgroups of patients with varying levels of comorbidity.

Second, multiple comparisons that consider variables other than TIBI score to define subgroups may affect the interpretation of P values. Third, our study includes only 1 sample of patients in 1 country with minimal ethnic and racial diversity, which may limit the generalizability of findings. Finally, we did not test other measures of comorbidity derived from other data sources (21), but we would expect them to produce similar results (22).

The results of our 5-year observational study suggest that any reduction of risk for cardiovascular events associated with tight glycemic control may not be uniform across patient subgroups. Only clinical trials that include relatively young and healthy diabetic patients can causally demonstrate that patients with low levels of comorbidity can benefit from attaining tight glycemic control. However, our study suggests that intensive glucose control may not have the expected protective effect on cardiovascular event risk for a substantial group of patients with type 2 diabetes who have high levels of comorbidity. Comorbidity may be an important consideration when tailoring glucoselowering therapy in patients with type 2 diabetes.

From the University of California Irvine, Irvine, California, and Consorzio Mario Negri Sud, Santa Maria Imbaro, Italy.

15 December 2009 Annals of Internal Medicine Volume 151 • Number 12 859

Grant Support: By Pfizer of Italy.

Potential Conflicts of Interest: None disclosed.

Reproducible Research Statement: *Study protocol and statistical code:* Available from Dr. Nicolucci (nicolucci@negrisud.it). *Data set:* Not available.

Requests for Single Reprints: Sheldon Greenfield, MD, Health Policy Research Institute, 100 Theory, Suite 110, Irvine, CA 92697; e-mail, sgreenfi@uci.edu.

Current author addresses and author contributions are available at www .annals.org.

References

1. Skyler JS, Bergenstal R, Bonow RO, Buse J, Deedwania P, Gale EA, et al; American Diabetes Association. Intensive glycemic control and the prevention of cardiovascular events: implications of the ACCORD, ADVANCE, and VA diabetes trials: a position statement of the American Diabetes Association and a scientific statement of the American College of Cardiology Foundation and the American Heart Association. Diabetes Care. 2009;32:187-92. [PMID: 19092168]

2. American Diabetes Association. Standards of medical care in diabetes—2008. Diabetes Care. 2008;31 Suppl 1:S12-54. [PMID: 18165335]

3. Qaseem A, Vijan S, Snow V, Cross JT, Weiss KB, Owens DK; Clinical Efficacy Assessment Subcommittee of the American College of Physicians. Glycemic control and type 2 diabetes mellitus: the optimal hemoglobin A1c targets. A guidance statement from the American College of Physicians. Ann Intern Med. 2007;147:417-22. [PMID: 17876024]

4. Patel A, MacMahon S, Chalmers J, Neal B, Billot L, Woodward M, et al; ADVANCE Collaborative Group. Intensive blood glucose control and vascular outcomes in patients with type 2 diabetes. N Engl J Med. 2008;358:2560-72. [PMID: 18539916]

5. Gerstein HC, Miller ME, Byington RP, Goff DC Jr, Bigger JT, Buse JB, et al; Action to Control Cardiovascular Risk in Diabetes Study Group. Effects of intensive glucose lowering in type 2 diabetes. N Engl J Med. 2008;358:2545-59. [PMID: 18539917]

Duckworth W, Abraira C, Moritz T, Reda D, Emanuele N, Reaven PD, et al; VADT Investigators. Glucose control and vascular complications in veterans with type 2 diabetes. N Engl J Med. 2009;360:129-39. [PMID: 19092145]
Kelly TN, Bazzano LA, Fonseca VA, Thethi TK, Reynolds K, He J. Systematic review: glucose control and cardiovascular disease in type 2 diabetes. Ann Intern Med. 2009;151:394-403. [PMID: 19620144]

8. Ray KK, Seshasai SR, Wijesuriya S, Sivakumaran R, Nethercott S, Preiss D, et al. Effect of intensive control of glucose on cardiovascular outcomes and death in patients with diabetes mellitus: a meta-analysis of randomised controlled trials. Lancet. 2009;373:1765-72. [PMID: 19465231]

9. Holman RR, Paul SK, Bethel MA, Matthews DR, Neil HA. 10-year follow-up of intensive glucose control in type 2 diabetes. N Engl J Med. 2008;

359:1577-89. [PMID: 18784090]

10. UK Prospective Diabetes Study (UKPDS) Group. Intensive blood-glucose control with sulphonylureas or insulin compared with conventional treatment and risk of complications in patients with type 2 diabetes (UKPDS 33). Lancet. 1998;352:837-53. [PMID: 9742976]

11. Huang ES, Zhang Q, Gandra N, Chin MH, Meltzer DO. The effect of comorbid illness and functional status on the expected benefits of intensive glucose control in older patients with type 2 diabetes: a decision analysis. Ann Intern Med. 2008;149:11-9. [PMID: 18591633]

12. Montori VM, Fernández-Balsells M. Glycemic control in type 2 diabetes: time for an evidence-based about-face? Ann Intern Med. 2009;150:803-8. [PMID: 19380837]

13. Singh S, Loke YK, Furberg CD. Inhaled anticholinergics and risk of major adverse cardiovascular events in patients with chronic obstructive pulmonary disease: a systematic review and meta-analysis. JAMA. 2008;300:1439-50. [PMID: 18812535]

14. Lee TA, Pickard AS, Au DH, Bartle B, Weiss KB. Risk for death associated with medications for recently diagnosed chronic obstructive pulmonary disease. Ann Intern Med. 2008;149:380-90. [PMID: 18794557]

15. Belfiglio M, De Berardis G, Franciosi M, Cavaliere D, Di Nardo B, Greenfield S, et al. QuED Study Group—quality of care and outcomes in type 2 diabetes. The relationship between physicians' self-reported target fasting blood glucose levels and metabolic control in type 2 diabetes. The QuED Study Group—quality of care and outcomes in type 2 diabetes. Diabetes Care. 2001; 24:423-9. [PMID: 11289462]

16. Franciosi M, Pellegrini F, De Berardis G, Belfiglio M, Cavaliere D, Di Nardo B, et al; QuED Study Group. The impact of blood glucose self-monitoring on metabolic control and quality of life in type 2 diabetic patients: an urgent need for better educational strategies. Diabetes Care. 2001;24:1870-7. [PMID: 11679449]

17. Pellegrini F, Belfiglio M, De Berardis G, Franciosi M, Di Nardo B, Greenfield S, et al; QuED Study Group. Role of organizational factors in poor blood pressure control in patients with type 2 diabetes: the QuED Study Group quality of care and outcomes in type 2 diabetes. Arch Intern Med. 2003;163:473-80. [PMID: 12588208]

18. Greenfield S, Sullivan L, Dukes KA, Silliman R, D'Agostino R, Kaplan SH. Development and testing of a new measure of case mix for use in office practice. Med Care. 1995;33:AS47-55. [PMID: 7723461]

19. Stier DM, Greenfield S, Lubeck DP, Dukes KA, Flanders SC, Henning JM, et al. Quantifying comorbidity in a disease-specific cohort: adaptation of the total illness burden index to prostate cancer. Urology. 1999;54:424-9. [PMID: 10475347]

20. Litwin MS, Greenfield S, Elkin EP, Lubeck DP, Broering JM, Kaplan SH. Assessment of prognosis with the total illness burden index for prostate cancer: aiding clinicians in treatment choice. Cancer. 2007;109:1777-83. [PMID: 17354226]

21. Charlson ME, Pompei P, Ales KL, MacKenzie CR. A new method of classifying prognostic comorbidity in longitudinal studies: development and validation. J Chronic Dis. 1987;40:373-83. [PMID: 3558716]

22. Corser W, Sikorskii A, Olomu A, Stommel M, Proden C, Holmes-Rovner M. Concordance between comorbidity data from patient self-report interviews and medical record documentation. BMC Health Serv Res. 2008;8:85. [PMID: 18416841]

Annals of Internal Medicine

Current Author Addresses: Drs. Greenfield, Billimek, and Kaplan: Health Policy Research Institute, 100 Theory, Suite 110, Irvine, CA 92697.

Mr. Pellegrini, Ms. Franciosi, Ms. De Berardis, and Dr. Nicolucci: Consorzio Mario Negro Sud, Via Nazionale, 8/A Santa Maria Imbaro, Italy.

Author Contributions: Conception and design: S. Greenfield, M. Franciosi, G. De Berardis, A. Nicolucci, S.H. Kaplan.

Analysis and interpretation of the data: S. Greenfield, J. Billimek, F. Pellegrini, M. Franciosi, G. De Berardis, A. Nicolucci, S.H. Kaplan. Drafting of the article: S. Greenfield, J. Billimek, F. Pellegrini, M. Franciosi, G. De Berardis, A. Nicolucci, S.H. Kaplan.

Critical revision of the article for important intellectual content: S. Greenfield, J. Billimek, M. Franciosi, G. De Berardis, A. Nicolucci, S.H. Kaplan.

Final approval of the article: S. Greenfield, J. Billimek, F. Pellegrini, M. Franciosi, G. De Berardis, A. Nicolucci, S.H. Kaplan.

Provision of study materials or patients: M. Franciosi, G. De Berardis, A. Nicolucci, S. Greenfield.

Statistical expertise: J. Billimek, F. Pellegrini, A. Nicolucci, S.H. Kaplan. Obtaining of funding: A. Nicolucci.

Administrative, technical, or logistic support: S. Greenfield, J. Billimek. Collection and assembly of data: M. Franciosi, G. De Berardis, A. Nicolucci.